Author's response to reviews

Title: An intron polymorphism of fibronectin gene is associated with end-stage osteoarthritis in a Han Chinese population: two-stage case-control study

Authors:

Hsin-Yi Yang (b262728@gmail.com)
Sui-Lung Su (a131419@gmail.com)
Yi-Jen Peng (yijen0426@gmail.com)
Chih-Chien Wang (tsghcc@yahoo.com.tw)
Herng-Sheng Lee (herngsheng131419@gmail.com)
Donald M Salter (Donald.Salter@ed.ac.uk)
Chian-Her Lee (chianherlee@yahoo.com.tw)

Version: 4 Date: 25 April 2014

Author's response to reviews: see over
Dear Dr Cornacchia,

I wish to thank the reviewers for their review of our manuscript entitled “An intron polymorphism of the fibronectin gene is associated with end-stage osteoarthritis in a Han Chinese population: two independent case-control studies”. We have responded to all of the comments raised by the reviewers, and our response can be seen in an accompanying document.

We added the e-mail addresses of all authors to the title page, and the entire manuscript was also re-edited by Edanz editing, as recommended by BioMed Central (Our Reference No: G1404-6091-Lee).

We hope that our manuscript is now suitable for publication in BMC Musculoskeletal Disorders.

Sincerely yours,

Chian-Her Lee, MD
Department of Orthopedics, School of Medicine, College of Medicine, Taipei Medical University and Hospital, Taiwan
E-mail: chianherlee@yahoo.com.tw